Author's response to reviews

Title: Mediators of the relation between War Experiences and Suicidal Ideation among Former Child Soldiers in Northern Uganda: The WAYS Study

Authors:

Kennedy Amone-P’Olak (kpamone@gmail.com)
Tlholego M Lekhutlile (Tlholego.Lekhutlile@mopipi.ub.bw)
Richard Meiser-Stedman (Richard.Meiser-Stedman@mrc-cbu.cam.ac.uk)
Emilio Ovuga (emilio.ovuga@gmail.com)

Version: 4 Date: 30 August 2014

Author's response to reviews: see over
August 30, 2014

The Editor in Chief
BMC Psychiatry

Dear Sir/Madam,

**MS: 1089182111125903 - Mediators of the relation between War Experiences and Suicidal Ideation among Former Child Soldiers in Northern Uganda: The WAYS Study**

Once again, I am delighted to resubmit our manuscript edited in response to editorial requests. We are extremely grateful that the manuscript has finally been accepted for publication in your esteemed journal, *BMC Psychiatry*. The following changes have been made to the manuscript:

1. The title page has been reformatted in accordance with the in-house style of BMC Psychiatry as requested by the editor and is now being submitted as: Title page KAP_R3. Particularly, the title of the manuscript, full list of names, institutional affiliations including email addresses of all authors, and corresponding author have been provided.

2. Figures 1 and 2 have been reformatted in accordance with editorial request. The figures have been reformatted portrait on PPT without the legends. The legends for figures 1 and 2 have been included immediately after the references on pages 22 and 23 respectively.

3. Additional files (i.e. tables) have been reformatted too in accordance with editorial request. The legends have been removed from the tables and included on pages 24, 25, and 26 for tables 1 through 3, respectively.

We are optimistic that, this manuscript has now been appropriately formatted and will be suitable for publication in your esteemed journal, *BMC Psychiatry* this time round.

Yours sincerely,

Dr. Kennedy Amone-P’Olak, Ph.D.
Associate Professor, Department of Psychology
University of Botswana